ARTICLE IN PRESS

The British Accounting Review xxx (xxxx) xxx



Contents lists available at ScienceDirect

The British Accounting Review

journal homepage: www.elsevier.com/locate/bar



The role and impact of professional accountancy associations on accounting education research: An international study

Angus Duff a, *, Phil Hancock b, Neil Marriott c

- ^a Accounting, Finance and Law, Business School, University of the West of Scotland, Ayr Campus, Ayr, KA8 0SX, UK
- ^b Faculty of Arts, Business, Law and Education 2017, The University of Western Australia (M404), 35Stirling Highway, PERTH, WA, 6009, Australia
- ^c Business School, University of Winchester, Winchester, SO22 4NR, UK

ARTICLE INFO

Article history: Received 6 July 2018 Received in revised form 6 March 2019 Accepted 16 March 2019 Available online xxx

Keywords:
Accounting academy
Accounting education
Accounting research
Institutional logics
Professional accountancy association

ABSTRACT

This paper explores the relative influence of the accounting academy and accountancy professional associations in the debate concerning the profile and quality of accounting education research. This research analyses 13 semi-structured interviews undertaken with members of key accounting professional associations in Australia, New Zealand (ANZ) and the United Kingdom and Ireland (UKI). The paper makes a theoretical contribution by the novel application of the institutional logics theory to the literature concerning professional accounting associations (PAAs) demonstrating the dominant commercial logic of the education function and the more traditional fiduciary logic of the technical function. The research finds that the primary stakeholders in the professional accounting curriculum development model are the PAAs and employers, whilst the accounting academy is relatively absent. The relative independence of the education and research and technical functions within PAAs is also identified: academic research and technical activity has little influence on professional education and vice versa. However, PAAs' funding of academic research is common across all four countries for brand recognition and in some instances to influence policy rather than informing the professional curriculum.

© 2019 Elsevier Ltd. All rights reserved.

1. Introduction

You don't want to frighten the horses ... ultimately, we have a commodity qualification which we want employers to take and train their staff inWe know that academics want to pursue academic journals ... but that's not what we're ... interested in. We want to see some outputs for practitioners and not academics ...(UKI2)

It has been long been recognised that research in accounting education is less well regarded compared to other areas of accounting research (Marriott, Stoner, Fogarty, & Sangster, 2014; Sangster, 2015) and this has raised concerns that this will have an adverse impact on the quality of accounting education research in the future (McGuigan, 2015). However, concerns about the quality of research published in mainstream accounting journals have also been identified (e.g., Basu, 2012; Demski, 2007; Hopwood, 2007; Moser, 2012). Problems identified suggest accounting research is stagnating (Moser, 2012), that it has little impact on professional practice (Basu, 2012; Demski, 2007; Hopwood, 2007; Kaplan, 2011) and heavily influenced by the

E-mail addresses: angus.duff@uws.ac.uk (A. Duff), phil.hancock@uwa.edu.au (P. Hancock), neil.marriott@winchester.ac.uk (N. Marriott).

https://doi.org/10.1016/j.bar.2019.03.004

0890-8389/© 2019 Elsevier Ltd. All rights reserved.

Corresponding author.

economics and finance literatures (Oler, Oler, & Skousen, 2010). For the purposes of this paper we define accounting research as the production of accounting knowledge, accounting education as the communication of accounting knowledge, and professional practice as the application of accounting knowledge. It is axiomatic that there is a recursive relationship between the three. For example, the application of knowledge (practice) naturally leads to the creation of new knowledge (research); the process of communicating knowledge (education) allows the deliverer and recipients to create their own interpretations and understandings (research); and engaging students in professional practice, e.g. as internships, is an education.

Yet accounting education special interest groups of academic accounting associations across the globe continue to have the largest membership base of all special interest groups. Accounting academics will often be members of their accounting or related special interest group, e.g. financial reporting, management control, public sector, auditing, etc., and will simultaneously be members of the accounting education special interest group of their region. This is because the vast majority of faculty members are expected to teach and there are benefits from the nexus that relates these two aspects of academic activity (Duff & Marriott, 2017a, 2017b; Hancock, Marriott, & Duff, 2017). Accounting education research which is generally thought to involve research into pedagogy and the teaching of accounting, cannot be disassociated from, or be less well-regarded than, mainstream accounting research. They are interrelated and necessarily so as they impact on the quality of the education experience of the students that will form the future membership and thinking of the accounting profession. However, accounting education research itself has not been without its critics. For example, Rebele, St, and Pierre (2015) argue that accounting education research has stagnated, mirroring concerns raised about the mainstream accounting literature. Specifically, Rebele and St. Pierre (2015 p.126) identify that the majority of accounting education papers are not empirical, relate to general topics and ignore issues relevant to accounting education practice.

A recent solution to the perceived problems of accounting education from the United States (US) was the publication of the Pathways Commission on Accounting Higher Education Report (Pathways) (2012). Pathways, a collaborative venture between the accounting industry and accounting academe, recommended transforming accounting into a learned profession that integrated accounting research, education and practice (Ratzinger-Sakel & Gray, 2015, p. 77). This paper queries how, in an increasingly globalised market for accounting education, professional accounting associations (PAAs) in the United Kingdom and Ireland (UKI) and Australia and New Zealand (ANZ) integrate accounting education research and practice. While an earlier paper by Thornton, Jones, and Kury (2005) examined professional accounting firms and applied institutional logics theory, the focus in this paper is on professional accounting associations. We believe our paper provides evidence that the move to a more commercial logic that Thornton et al. (2005) demonstrated applies to accounting firms also applies to PAAs.

A touchstone of the professionalism project are professions' embedded and deep relationship with universities and higher education (Abbott, 1988; Friedson, 1994; Larson, 1977; Wilkerson, 2010). Fundamentally, the literature describing relations between education and professionalism reports an almost symbiotic relationship between professions and universities (Evans, 2010). However, within the accounting profession this strong relationship has been challenged by the growth of PAAs training non-relevant graduates and the introduction and growth of professional accounting trainees hired directly from school.

PAAs have developed longstanding relationships with universities in four key ways. First, by the accreditation of university programmes, providing degrees of assurance to potential students of their content and the future employability of graduates and exemption from the qualifying examinations of the PAA. Second, by the provision of research funding that covers the marginal costs of research that may be of some value to the PAA, for example in that the research chimes with a policy agenda within the association or to promote their commercial objectives. Third, they also support professional development activities which assist the academic in the classroom. Finally, they undertake thought leadership activities, opinion pieces that sometimes engage academic and professional audiences.

This paper considers the relationship between the education and technical or research functions of PAAs in ANZ and the UKI. The education function is generally responsible for the education requirements for becoming a member and then to provide opportunities for continuing professional development. The CA ANZ web site includes details how to "develop the skills you need to thrive in the changing business world with our range of learning opportunities and tools." (https://www.charteredaccountantsanz.com/learning-and-events/learning). The technical function seeks to "actively engage with the government, regulators and standard-setters to advocate on behalf of the accounting profession." (https://www.charteredaccountantsanz.com/member-services/technical). The UKI countries are chosen as they represent the birthplace of PAAs and also because they have six PAAs with Royal Charter status. ANZ is chosen as a setting, with well-developed PAAs, where UKI-based PAAs also operate. Historically ANZ has substantial linkages to the UK, including its membership in the Commonwealth. We argue that PAAs have not necessarily significantly influenced the quality of accounting education research despite their role in influencing the accounting curriculum taught in higher education and offering incentives to pursue lines of research using favoured methodologies. This paper provides a novel investigation into the quality of accounting education research by considering the roles and relationships between two functions within a PAA that intersect with the accounting academy.

¹ Institute of Chartered Accountants of England and Wales, (ICAEW), Association of Certified Chartered Accountants (ACCA), Chartered Institute of Management Accountants (CIMA), Chartered Institute of Public Finance Accountants (CIPFA), Institute of Chartered Accountants of Scotland (ICAS) and Institute of Chartered Accountants of Ireland (ICAI).

The aim of this research is to consider the interaction of the PAAs with the accounting academy in relation to accounting education research. Specifically, the objectives of the research are three-fold. The first is to consider the roles of the education and technical departments of PAAs. Secondly, we consider how PAAs' education and technical departments interact with each other and, in turn, with accounting academia. Finally, we identify and document the dominant institutional logics of technical and education departments of PAAs.

This research draws on an analysis of 13 semi-structured interviews undertaken as part of a larger study with key organisational actors responsible for education and research within PAAs in ANZ and the UKI. The interview evidence is supplemented with archival evidence concerning PAAs available in the public domain. To analyse our findings, we adopt an institutional logics approach (Friedland & Alford, 1991) to consider the dynamics of change within the PAAs' education and technical functions. In doing so, we view the PAAs themselves, their relations with academe, and their internal relations as dynamic and in a state of flux, rather than static and stable.

The contribution of this paper lies in four areas. First, the paper makes a theoretical contribution by the novel application of the institutional logics theory to the literature concerning PAAs, demonstrating the dominant commercial logic of the education function and the more prevalent fiduciary logic of the technical function. Institutional logics has been recently applied to professional service firms offering accountancy-related services (e.g. Lander, Koene, & Linssen, 2013; Suddaby, Saxton, & Gunz, 2015; Thornton et al., 2005). Significantly, in some instances, these associations have moved beyond their traditional national boundaries to create ambitious global organisations. Second, is the demonstration that in the development of the accounting curriculum, the primary stakeholders are PAAs and employers, where the voices of academy are relatively silent. PAAs' aim to create generic qualifications that 'future proof' the employability of their newly-qualified members. Third, is the identification of the independence of the education and research and technical functions within, where research and technical activities of PAAs has relatively little influence on accounting education and vice versa. Research activity within PAAs is frequently aimed at developing brand awareness, rather than the relevance and efficacy of the qualification. The exclusion of research from the qualification means that it focuses on professional practice (practitioners), rather than research (academics). The detachment of research from education is at the heart of concerns about the quality of accounting education research. We suggest this is part of the 'accounting education research stagnation' problem identified by Rebele and St. Pierre (2015, p.128). Fourth, to contribute to the slender corpus of literature on PAAs in a post-professional model (Suddaby, Gendron, & Lam, 2009) or period of 'deprofessionalization' (Lampe, Garcia, & Tassin, 2016, p. 3).

Finally, the paper concludes that PAAs have a role in promoting the impact and value of accounting education research by explicitly incorporating pedagogic research, including professional curriculum development, in their research strategies. Such pedagogic research could include the development of, and interaction between, the university and professional curriculum. University educators have long recognised the role of degree level study in developing transferable skills and a life-long approach to learning. For example, it is important to teach why accounting standards are needed and how they are developed and not to just focus on the content of the accounting standards of the day.

This paper proceeds as follows. Section two provides a brief discussion on the role of PAAs and an overview of the accountancy profession in ANZ and in the UKI. The literature review that follows in section three considers the development of the accountancy profession along with the institutional logics theory which forms the theoretical model for the analysis. The design, conduct and analysis of the semi-structured interviews that constitute the primary research method are considered in section four. Section five reports the findings. Section six provides a discussion of the findings and the final section concludes the paper.

2. Professional accountancy associations in Australia, Ireland, New Zealand and the United Kingdom

2.1. Professional accountancy associations in Australia, Ireland, New Zealand and the United Kingdom and educating accountants

As stated earlier the paper looks at PAAs in UKI and ANZ. The UKI countries as it represents the birthplace of PAAs and also because they have six PAAs with Royal Charter status. ANZ is chosen as there are also well-developed PAAs, where UKI-based PAAs also operate. Historically, ANZ has substantial linkages to the UK, including its membership of the Commonwealth. Table 1 identifies the names of the PAAs included in this study. According to the Financial Reporting Council (2017), six PAAs report

Table 1 Professional associations included in the study.

PAA	Member numbers
CPA Australia	165,000
Chartered Accountants Australia and New Zealand	110,000
Association of Chartered Certified Accountants	200,000
Institute of Chartered Accountants in England and Wales	150,000
Institute of Chartered Accountants in Scotland	21,500
Institute of Chartered Accountants in Ireland	26,000
Chartered Institute of Management Accountants	150,000
Chartered Institute of Public Finance and Accountancy	14,000

some 350,000 members are in the UKI and over 515,000 worldwide as at 31st July 2017. In ANZ there are three professional accountancy associations: CAANZ reports 117,000 members (CAANZ, 2018); CPA Australia (CPA) identifies a global membership of more than 165,000 members (CPA, 2018); and the Institute of Public Accountants records over 35,000 members (IPA, 2018).

Historically, the education of accountants has long been a contested territory. Carr-Saunders and Wilson (1933, p.223) note:

'The present system of education', writes a Chartered Accountant of some standing, 'is far from satisfactory.' The universities play no part in the scheme, except in Scotland, except that graduates are exempt two years articles; and the dearth of proper places of instruction makes the accountancy profession the happy hunting ground of coaching institutions.

In some ways then, the problems identified eighty five years ago still persist. Much professional accounting education required by accountancy 'qualifying bodies' (Millerson, 1964) in the UK is delivered outside universities by private training companies and to increasing numbers of non-relevant graduates: a stark contrast to most other Global Accounting Alliance countries. Notably both Larson (1977) and Abbott (1988) identify the pivotal role education plays in the professionalization project's ultimate goal of achieving social respectability.

Gammie and Kirkham (2008) document the Institute of Chartered Accountants of Scotland's (ICAS)' move to re-open a school-leaver entry route, previously closed by the Institute as a consequence of the Charles Report (1970). Indeed, Gammie and Kirkham (2008, p.357) suggest it could be viewed as 'a retrograde step for an Institute wishing to position itself at the forefront of the accountancy profession' and identify the motives behind the recent move were primarily commercial in protecting its future income stream (p.358). By contrast, Annisette and Kirkham (2007), in an analysis of the English chartered accountancy profession with its relatively low intake of accountancy graduates, view the separation of academy and practice as mutually beneficial to universities and the profession. The profession benefits from universities as a major user of graduate talent, while university departments of accountancy remain relatively free from the constraints of accreditation. With doctoral-qualified faculty, accounting departments have multiplied their research efforts over the past four decades. In turn, the profession benefits from the significantly higher status that has accrued to accounting academe as a consequence of its own significant educational development since the 1980s (Annisette & Kirkham, 2007).

Traditionally the three UK institutes (Institute of Chartered Accountants in England and Wales, ICAEW; in Ireland CAI; and of Scotland, ICAS) that focus on public practice work, undertaken by accountancy firms, have recruited mostly graduates. The residual three institutes, the Association of Chartered Certified Accountants (ACCA), the Chartered Institute of Management Accountants (CIMA), and the Chartered Institute of Public Finance and Accounting (CIPFA), have always maintained a non-graduate entry route. However, UK accountancy firms are increasingly recruiting school-leavers at the expense of graduates. This is for two reasons: first, to reduce training costs, as graduates command a salary premium; second, in the name of 'business stability' as school-leavers take typically five years to train compared to three for graduates, reducing staff turnover; and third, concerns over mounting student debt, deterring some individuals from entering higher education (Duff, 2017). In ANZ most entrants to a PAA complete a tertiary qualification before completing the Professional Level exams and required practical experience. However, there are many private providers in ANZ who are able to provide the required degree and preparation at significantly lower costs as they do not have to support a research function as do universities.

So, in both the UKI and ANZ, the PAAs are responsible for the professional curriculum that is followed by their student members. Students follow a training programme taught while in practice or by completing an approved programme of study at university that prepares graduates for student membership of various PAAs. Consequently, the development and content of the professional curriculum is an important role for the PAAs which are all members of the International Federation of Accountants (IFAC) comprising 175 members in more than 130 countries representing over 3 million accountants. IFAC sets education standards for all members through the International Accounting Education Standards Board (IAESB). International Accounting Education Standard (IES) 1 (IFAC, 2017) requires all PAAs to set entry requirements so that applicants have a reasonable chance of passing the requirements to become a professional accountant. Hence, the PAAs have a responsibility for ensuring the entry requirements such as a degree as required with all the major PAAs in the UKI and ANZ allow a reasonable chance of passing their professional level examinations. The next section provides a review of the literature on professional associations and introduces Institutional Logics theory used in this paper.

3. Literature review

3.1. 1 historical perspectives on the development of the accountancy profession

Investigations of professions (e.g. engineering, law, medicine) generally point to a number of defining traits, as sociological studies of professions tend to eschew a precise definition (Carnegie, Edwards, & West, 2003). For example: a body of knowledge; the ability to self-regulate and discipline; the existence of a code of ethics, a relationship with clients based on trust, the belief that members act as professionals; and the ability to secure an enhanced remuneration as a consequence of professional status (e.g., Carr-Saunders & Wilson, 1933; Goode, 1961; Greenwood, 1957; Larson, 1977; Macdonald, 1995; Millerson, 1964). Alongside these defining traits traditional understandings of professional work are that it is borne out of

sense of social duty beyond the remuneration that accrues to professional employment (Carr-Saunders & Wilson, 1933; Lee, 1995).

The development of public accountancy profession has been the subject of continued scholarly interest. Lee's (1995) study of the professionalization of accountancy is seminal in this regard documenting the rise of professional societies of accountants to provide market control of accountancy services. As mentioned earlier, the Pathways (2012) report recommended transforming accounting into a learned profession that integrated accounting research, education, and practice. To understand the development of the accountancy profession, Richardson's (2017) stylized history is considered. Six stages sequentially are identified: (i) the emergence of the profession; (ii) the closure of the profession; (iii) engagement with the state and the embedding of accounting expertise in regulation; (iv) globalization; (v) commercialization; and (vi) the creation of hybrid institutions with emergent features. For the purposes of this research we focus on the final three stages of globalization, commercialization and creation of hybrid institutions with emergent features.

The globalisation of the accountancy profession is characterised by three waves of activity Richardson (2018). First, was the creation of imitation professional associations across the British Empire, often carrying the term 'Chartered' but without corresponding royal ascent and different educational requirements. Second was the creation of a branch structure for some UK-based associations such as ACCA and the Chartered Institute of Management Accountants (CIMA) in establishing global associations with common educational programmes. Consequently, overseas members acquire the same degree of legitimacy as members in the UK. Third, other national associations developed reciprocity agreements with other members of sister associations, e.g., the Global Accounting Alliance of Institutes of Chartered Accountants of England and Wales, of Scotland, in Ireland and other former Empire nations. Pressures to harmonize the accounting profession are evident in the work of the International Accounting Standards Board (IASB) and the International Federation of Accountants (IFAC).

Following on from globalization, the accountancy professionalization project is characterised by a process of commercialisation. Richardson (1997, p.635) argues that the professionalization of accounting has at best been partial. This reflects the inability of the profession to gain statutory recognition for their occupational domain, a non-standard basis for the education of their practitioners, and market dynamics have shifted away from the traditional core service lines of audit and accounting. In particular, the accountancy marketplace is said to have be too dynamic to allow the professionalization of accountancy to mature (Richardson, 1997). Furthermore, the profession itself has moved away from being self-regulating to being regulated (e.g. Sarbannes-Oxley (SOX) enacted in the US in 2002), in the wake of a storm of financial scandals at the turn of the 21st century (see Lampe et al., 2016). Accountancy firms have been eager to develop new service lines away from traditional areas of activity. These 'professional service firms' no longer use the title 'Chartered Accountants' in their names and the core service line of audit is now often just a small fraction of their fee income. Consequently, firms have moved from a professional model to a commercial model (Hanlon, 1994, 1997).

Finally, reflecting the present state of the accountancy profession, is that of the movement to a post-professional model (Suddaby, Cooper, & Greenwood, 2007). In the wake of financial scandals and formal regulation of accountancy work, the accountancy profession has been described as directionless and pursuing a variety of strategies, sometimes described as a process of 'deprofessionalisation' (Lampe et al., 2016 p.3). Other scholars suggest the profession is creating new hybrid forms described by commercial and professional logics (Richardson, 2017). Alongside these developments there remain tensions between the local and the global. Major accountancy practices operate on a global scale, within a framework of the IASBs international accounting standards and international professional federations such as IFAC, yet local PAAs continue to thrive (Ramirez, 2009). However, recent literature focuses on the role of global accountancy practices; a much slender corpus of work considers the relevance of local models of PAAs. The next section describes institutional logics as the theoretical lens adopted for this investigation.

3.2. Institutional logics theory

Institutional logics theory (Friedland & Alford, 1991) proposes that modern neo-liberal society is said to have central institutions each of which has incompatible and competing logics which provide a catalyst for change. Thus, the emphasis is on these differentiated competing institutional logics, rather than isomorphism. The institution itself has a central logic that forms its organising practices and symbols which are illustrated with vocabularies and self-identity (Friedland & Alford, 1991). In simple terms, an institutional logic explains how a social world works (Jackall, 1988). Institutional logics link the focus on macro-structures and cultural norms of isomorphism with agency perspectives where individual actors compete for resources within a contested field. In particular, institutional logics focus on the effect by which individuals organise themselves, make sense of their social reality, and (re)produce their subsistence (Ocasio & Thornton, 1999; Thornton & Ocasio, 1999). Thus, individual agency is emphasised.

More recent work in the field of accountancy has documented the shift from a fiduciary logic, emphasising a kindred sense of membership of a profession and acting in the public interest to a commercial logic, recognising the desire to generate income and surpluses and be seen as entrepreneur and member of a business elite. Naturally, as for-profit enterprises a tension has always existed between the two logics. However, the growing importance of the commercial logic at the expense of the professional logic has been noted in the past two decades (Lander et al., 2013; Suddaby & Greenwood, 2005). This tension between logics and organisational response is "important because they can have major implications for social legitimacy and thus an organization's access to critical resources. Organizational survival may even be at stake." (Greenwood, Raynard, Kodeih, Micelotta, & Lounsbury, 2011 p319).

Traditionally, accountancy operated a fiduciary logic whereby it sought equal status with other professions with a public interest mission, a responsibility to wider society rather than simply supplying services for profit (Jones, 1995). This was achieved, in part, by Victorian pioneers, such as Edwin Waterhouse, a founder of what would become the largest global audit firm, PWC. Notably, Waterhouse drew on his Quaker faith to establish a firm that promoted the value of integrity and independence (Thornton et al., 2005), principles of considerable value where financial statements were the primary tool available to investors. The founding fathers of accounting practice also operated a commercial logic, in that firms with a superior reputation were able to charge premium fees for their services, reputation being essentially an economic concept in institutional theory terms. However, in more recent times the commercial logic has evolved to resemble a more corporate form as demonstrated in Wm. R. Gregory's 1980 address to the American Institute of Certified Public Accountants' (AICPA) Public Oversight Board:

It seems that the effects of the phenomenal growth in the profession and competitive pressures have created in some CPAs attitudes that are intensely commercial and nearly devoid of the high-principled conduct that we have come to expect of a true professional. It is sad that we seem to have become a breed of highly skilled technicians and businessmen, but have subordinated courtesy, mutual respect, self-restraint, and fairness for a quest for firm growth and a preoccupation with the bottom line. (quoted in Zeff, 2003b, p. 267, p. 267)

In terms of market triggers, the profession was itself acting as a legitimacy agent; by its audit and attestation of companies' financial statements it legitimised these accounts.

Thornton et al, (2015) identify four dimensions that describe the mechanics of institutional and organisational change. These four dimensions are: mission conflicts; market triggers; governance forms; and changes in dominant institutional logics. The operationalisation of these four dimensions is described in column one of Table 2, using the literature describing the development of the accountancy profession in fiduciary and commercial logics.

This shifting sand in logics has been accompanied by a number of significant trends. These include: a desire to promote a wide and ever-changing range of services to their clienteles (Hopwood, 2008); the development of formal regulation of accounting services worldwide (e.g., Sarbanes Oxley in the United States); regulatory inspection of audit provision (e.g., the audit inspection unit in the UK); the growth of limited liability partnerships and non-partnership organisations providing services (Fletcher, Frecknall-Hughes, & Williams, 2013) and competition from other providers of accountancy services (e.g., sustainability audits provided by engineering consultants, financial advice provide by financial planners in Australia).

Table 2Four dimensions of institutional and organisations change — the traditional and contemporary environments.

Dimension	Traditional environment	References	Contemporary environment
1. Mission conflicts	Fiduciary logic: Build legitimacy of profession Commercial logic: Establish reputation of association and members	Dennis, 2000; Jones, 1995; Zeff, 2003a; Zeff, 2003a Walker, 1995, 1999, 2004	Fiduciary logic: Maintain market position of profession against competitors Commercial logic: Membership growth, education and CPD services 'adding value to membership' Only research relevant to practitioners is highly valued.
2. Market triggers	Fiduciary logic: Sell legitimacy Promote educational attainment of members Commercial logic: Jurisdiction over audit and tax services Development of other service lines	Van Wyhe, 2007 Dennis (2000) Richardson (1988) Thornton et al. (2005) Previts (1985)	Fiduciary logic: Globalization, international legitimacy of members and practitioners Managing expectations of evaluating audiences Commercial logic: Growth through mergers and networks of similar associations Provision of range of services to members (customers) Benchmarking against competing associations
3. Governance forms	Fiduciary logic: Relevance of profession and addressing public interest issues Commercial logic: Development of commercial member services and training to non-members	Lee (1995) Zeff (2003b) Greenwood, Suddaby, and Hinings (2002)	Fiduciary logic: Relevance of profession and addressing public interest issues Commercial logic: Development of the brand, including internal brands (e.g. faculties)
4. Changes in dominant logics	Fiduciary logic: Relevance of profession and addressing public interest issues Relationship with universities for legitimacy reasons Commercial logic: Competition between PAAs	Jones (1995) Annisette and Kirkham (2007) Armstrong (1985) Richardson (2018)	Fiduciary logic: Decoupling of relations with universities, in some cases direct competition Reduction in university accreditation Redefinition of public interest duty Commercial logic: Significant growth in membership numbers Selling of additional CPD to members Creation of PAA brands

Furthermore, the development of international financial reporting standards has required firms to have operations in multiple jurisdictions. In the next section the research method is explained.

4. Research method

In accordance with the exploratory nature of this paper, the emergent state of knowledge in this domain, and prior critical interpretative literature, field research is most suited to address the research questions. Accordingly, the research uses a qualitative design and the appropriate ethics approvals were obtained. The choice of a semi-structured interview as the principal method of data collection is justified by the exploratory nature of the study being undertaken (Silverman, 2014).

Thirteen semi-structured interviews were conducted with senior personnel responsible for 13 education and technical representatives of PAAs. Eight of these representatives were based in the UKI and five in ANZ. The aim of these interviews was to ascertain the linkages between the education and research functions within their organisations and their modus operandi. Interviews were arranged by contacting the PAA directly.

Given the small sample size and the seniority of the interviewees they will only be identified as ANZ1 to 5 or UKI 1 to 8 to assist their anonymity. All interviews were conducted in the interviewees' own offices. Interviewees were initially contacted by telephone which was followed up by email. The significance of the research to potential participants was enhanced by sponsorship from both an ANZA and UK PAAs which provided funding for the projects and provided some publicity for the project on its website.

Interviews were typically of 60–90 min duration. Each interview was recorded using a digital dictation device and subsequently transcribed by an independent expert transcriber. The transcription was shared with the interviewee to enable any typographical errors to be corrected or commercially sensitive information to be redacted. Before the start of each interview a statement of the objectives of the project and a list of possible discussion points and questions were given to the interviewee. Informed consent was obtained from all participants prior to their interviews. The interviewees were guaranteed anonymity and they are only identified by the interview number and the location of their organisation. No further information is provided as given the size of the sample and the seniority of those involved it would be possible to ascertain the respondents' identities and compromise their anonymity.

The data analysis adopts the three processes identified by Saldaña (2009) of: first cycle coding, whereby the initial data is reduced; second cycle coding, involving a process of reorganising and analysing first-cycle data; and post-coding where conclusions are drawn and findings compared to prior literature allowing an assessment of the fit to the theoretical foundations of the research. The first cycle coding involves careful reading of each transcript along with the preparation of an analytic memo or reflective log of initial thoughts on the data and the participants. The transcripts are then re-read to develop a number of themes based on the data. NVivo was also used to assist in the first cycle coding. Second cycle coding allowed a degree of re-organising and re-analysing the data coded using first cycle methods (Saldaña, 2009). Re-reading the transcripts then provided a sense that a conceptual organization of the first cycle codes had been achieved, akin to a point of saturation where no new codes emerged. At this stage, linkages between the principal themes were sought (Creswell, 1994). The third cycle involved a process of relating these themes to the forms of institutional logics. Only a selection of quotations has been included to represent the views expressed. Often the sentiments expressed in these extracts are replicated in more than one interview. Where a sole view is expressed this is noted. In the next session the findings are presented.

5. Findings

The objectives of the research were to consider the roles of the education and technical departments of PAAs; their interaction with each other; their interaction with accounting academe; and the institutional logics that underpin their actions. The changing higher education logistics in the accounting discipline are analysed according to four dimensions identified by Thornton et al. (2012): mission conflicts, market triggers, governance forms, and changes in dominant institutional logics. The themes identified in the qualitative analysis of the interview data were arranged into the four dimensions and are shown in column four of Table 2.

5.1. Mission conflicts

Clinebell and Clinebell (2008) argue that originally business schools were criticised as being trade schools and they reacted by emphasising the academic side of business education. However, a recent shift in direction to include more real-world experience is now evident with business practitioners frequently found delivering teaching in universities. Paisey and Paisey (2017) frame this dimension as academic rigour giving way to a more corporate facing environment portraying more of a professional rigour.

There was evidence that like the professional accounting firms the PAAs have also adopted a more commercial logic in parts of the operations with a focus on education.

I do think there is a role for professional bodies to be commercial. That is not necessarily to make a profit, but to make a profit that they can reinvest back into adding value to its membership. It can do that in a number of ways, it can do that by growing its members, it can do that through its educational offerings and the seminars and events the like

that it can put on and charge a fee for. It can do it in a range of ways but I don't believe that it's well in the profit of lining the pockets of individuals or any organisation. (ANZ3)

It was evident that the PAAs each had a strong commercial focus evidenced by targets for increasing student registrations who would eventually become full members, creating an effective annuity of the PAA. For example:

I mean it is difficult because ... our own targets internally are all centred around student growth and student numbers and member growth and trying to bring more and more people on board, and therefore we have to be led by the employers and business and be market led in that respect. (UKI3).

There was also evidence that the professional bodies only valued academic research that was relevant or useful to the accounting practitioner and discounted the academic side of the research.

You know, something very obscure or esoteric wouldn't be something we would necessarily wish to put our name to, but if it's something that members would say "Oh, what does this tell us?" ... funding in hospitals and the health sector and health sector accounting, we've done some work on that, publications on that, (UKI1).

5.2. Market triggers

The triggers that cause institutional change and instability are one of Thornton et al.'s (2005) four dimensions. Our findings identify six market triggers. The first trigger identified in the interview data was *benchmarking* both in terms of monitoring other professional body activity and behaviours and measuring the commercial attractiveness of their qualifications.

 \dots the institute will give a lot of feedback from what's happened internationally \dots we would \dots monitor that. We also monitor what's happening in our sister institutes \dots we work very closely with them. (UKI1)

... we're always benchmarking our qualification with employers saying like "Is it fit for purpose? Does it meet your needs? Does it reflect what they're actually doing in the workplace? Does it equip them to do things in the workplace?" (UKI2)

The second trigger identified was the constant *need to promote their PAA brand*, not just in terms of marketing and public relations exposure for brand recognition and promotion, but positioning their brand as a unique international and influential thought leader. This was one of the main reasons for funding academic research and accrediting their qualifications in higher education institutions.

I think, in part, it is getting the brand out there to be perfectly honest. (ANZ4).

I think one blunt reason for funding academic research is that the professional bodies want to form a good relationship with academics for marketing because the more that academics value a particular professional body, the more likely they are to speak well of them in the classroom, the more well informed about that professional body students will be when they finish. So any professional body will want to get good brand recognition amongst the academic community and can use research funding to do that. So, on a more mercenary side, I would say that would be one reason for doing it. (UKI4).

The third market trigger identified came in the form of *reacting to competition from other providers*, whether this was in attempting to measure and maintain market share in key industries or countries or in an attempt to attract the limited supply of intellectual talent that entered the market.

Well, we wouldn't be directly competing with [Professional Body Name Removed]. [Professional Body Name Removed]. We would be instead competing directly with [Professional Body Name Removed]. [Professional Body Name Removed] would consider itself to be in competition here. I would say in reality [Professional Body Name Removed] ... has become very much more niche rather than ... It's marketing very hard, but it sits very much in that space. But [Professional Body Name Removed] very much would dominate the 28 year-olds to a certain extent that have discovered accounting and would like to do accounting. We would generally see we have a very strong position in relation to graduates coming out from university who go directly into training. (UKI1)

... we have a commodity qualification which we want employers to take and train their staff in and there are other competing qualifications out there. If we come up with something that's not recognised as relevant to the needs of the employer, they're going to say "Why the hell have you got that in there?" (UKI2)

A fourth market trigger was the need for the professional body to *demonstrate value to its membership*. This took various forms including using research funding to demonstrate an investment in the future of the professional body while recognising that it had to be seen as valuable in the practical environment in which their members practised.

... innovation and development was seen as quite silo-ed and the value of the research wasn't being recognised by the rest of the organisation. It was being recognised by our members and academics and other external stakeholders, but within the organisation it was difficult for other colleagues to see where all of this money that was being spent added value ... We reject [research] proposals that don't have a very clear benefit to our members ... It's almost the most common grounds for not even progressing it through a review stage. (UKI3)

I do think there is a role for professional bodies to be commercial ... to make a profit that they can reinvest back into adding value to its membership. (ANZ3).

It does remain a challenge though in that practitioners are continually questioning the value of putting theory into the formal programme itself. (ANZ4)

There was evidence of a desire to promote a range of services to a range of clientele, another market trigger. These services ranged from offering seminars and events, or the dissemination of research findings in readily accessible and digestible formats.

Our research reports were hard copy and available for sale only at £20 a copy. We are now making all of those research reports available as a [free] downloadable PDF as well ... So there's nothing now to stop people getting hold of them very easily. (UKI5)

Some of the work that we're doing ... at the moment we are producing pieces of work because we think that to improve financial management in developing countries these pieces of work need to be done if we're going to be able to move it forward. So, you know, it might be something that we think is important and needs doing and so we will use research to do it. (UKI4)

So [Name of Professional Body] will disseminate via its newsletters, publications and the events that it holds ... We have our own journal called [Name of Professional Body], which is for members, and so we will place things in there to disseminate to members. (UKI4)

A fifth market trigger was attempting to *meet a diverse range of stakeholders' expectations*. These main stakeholders included members, employers and students. The academic community was also recognised as a stakeholder, but their decisions often mediated or controlled, most often through the accreditation process. In terms of research, the PABs did not differentiate between accounting education research and other research due to the diversity of their curriculum while others specifically mentioned the value of pedagogic research especially technology enhanced learning initiatives and changes in assessment strategies.

I think our curriculum is very wide dealing with everything from accounting, auditing, corporate finance, ethics, governance, management etc. and the very wide work that goes on throughout all of those themes is part of our curriculum. So I consider accounting education research is actually all accounting research and indeed some management research because our curriculum is extremely wide. (UKI7)

[Name of University] had done some research looking at how to use digital technologies in the classroom ... what it developed into was here's a user-friendly digital resource that we can make freely available to accounting educators so that they can now see how digital technologies could be used in their learning processes. (ANZ3)

there was a research paper done for IFAC Education Board ... which we considered when ... updating our syllabus and one of the outcomes of that was to recommend in terms of assessment that the best way to train chartered accountants was to have them do effectively a series of mini projects and to use that as a basis for assessment as opposed to a [terminal] examination. (UKI1)

I think within the organisation there are different divisions and different units that look after the different stakeholder groups and so it very much mirrors the interests and so it is just constantly querying among the people who are working in those different divisions, how are we balancing the sometimes competing interests that we represent. (ANZ5)

The final market trigger came in the form of the *supply or availability of resources*. Specifically the professional curriculum and research funding were limited by availability of time, standards, human and financial resources.

We try to include content that is supported by research but, in terms of the constraints that we have to operate within, firstly we have quite discreet study periods so ... we have to then look at what are the most critical pieces in that subject area that they must know: how do we teach those: how do we structure that? (ANZ1)

You know, because when you think about the actual hours that you have, the teaching hours aren't that much really ... and then you're fitting it in around a job as well ... (UKI6)

We do have a fixed pot which is I will say (because we are getting a bit more open about what the fixed pot is) it is around a quarter of a million dollars. So those proposals then need to fit within the size of the pot each year. (UKI6)

5.3. Governance forms

The third dimension identified by Thornton et al. (2005) is *governance forms*. In all but one PAA, research and education were structurally separate and linkages between the two were generally scant and serendipitous.

my understanding is that we have a research strategy which is independently created, but is then tested against the technical policy board of the Institute which is one of the operational boards of our council. So the council sets a strategy, technical policy board flows from that and the research committee sit underneath that and we have, I guess, a limited number of areas that we could choose to work in and we try to limit it to 5 or 6 key themes and we seek to run with those

for 3 or 4 years. Whether any of that research then ends up being relevant to the education agenda within [name of PAA] tends to be accidental rather than deliberate because education doesn't particularly have an input into the way that our research strategy is set. So it may well be that some material that comes out of what [name of PAA] funds can be very relevant to education; equally well it may not be because of the nature of the work that's being done.(UKI7)

Research and development and education seemed to be disconnected in their scope which was exacerbated by structural tensions.

Because the professional body, this one, has effectively its own research arm, which is the technical department, and it goes away and it does research and it's about excellence, it's about, you know, advancing the science of accountancy and all the rest of it. I think that area feels that that's what research is about. It isn't about learning, it isn't about development; it's about the science of accountancy and how we can advance that and it's our understanding of it. It's cutting edge, blue sky thinking and that's the place for research and that's their view, whereas I would see it as much more applied. I'm a sort of dirty, oily mechanic. (UKI2)

This identified the tension between research, with its underlying fiduciary logic 'advancing the science of accountancy' and the 'applied' commercial nature of education, with its 'dirty oily mechanic' leader's quest for pragmatism.

However in one PAA, research and development was integrated with curriculum development, as a means of justifying expenditure on research. That is by linking research and development to the curriculum, added value was demonstrated.

Until this month, in fact, research sat within the [name of department], which was actually a sub-department of [name of department]. Now research has moved under education, under the directorship of [name], and so now research sits alongside the colleagues who are tasked with updating the curriculum, continuous professional development and, you know, meeting the training needs of our members, the students. (UKI3)

5.4. Changes in dominant institutional logics

Of the four dimensions it is the changes in dominant institutional logics that are potentially the most critical and powerful mechanisms of institutional and organisational change and stability (Paisey & Paisey, 2017; Thornton et al., 2005). The significant growth in membership numbers has been a characteristic of PAA s in the past 15–20 years. Some like ACCA and CPA Australia have increased numbers through growth into other jurisdictions (cf. Richardson, 2017) while others have grown through mergers like the Institute of Chartered Accountants in Australia and the New Zealand Institute of Chartered Accountants. For example:

We have 26,000 enrolments every semester (ANZ1)

As discussed earlier in this paper, more recent work in the field of accountancy has documented the shift from a professional fiduciary logic, emphasising a kindred sense of membership of a profession and acting in the public interest to a commercial logic, recognising the desire to generate income and surpluses and be seen as entrepreneur and member of a business elite. This tension between logics and organisational response is "important because they can have major implications for social legitimacy and thus an organization's access to critical resources. Organizational survival may even be at stake" (Greenwood et al., 2011 p319):

I think it's interesting times in Australia for professional bodies. I think all bodies will be looking towards their future and how they invest that value for members. I think there is potentially a risk of having a pendulum swing too far back in the direction where there is discounts given and not to say that these things aren't good things, but I don't know. Other than that I think they need to take the time to be very clear and balanced in the directions they take and don't swing too far back from the perspective of undervaluing what they provide by undercharging. Charges communicate quality and premium offering. It's not to say there shouldn't be free events or free offerings you know members pay their fees and they have a right to expect good things from it. (ANZ3)

The growth in size of PAAs has been consistently pursued by the major PAAs included in this study. Again this is consistent with the commercial logics being the more dominant logic:

Our international development work, Council has approved a strategy that that should be break even over the medium-term and therefore any gains that we make on consultancy work — so if we do tender for projects where there's a profit element and that might be consulting in the developed world — can be used and indeed have to be used for work where we perhaps don't receive a fee at all. (UKI7)

In the next section we discuss the findings before drawing some conclusions and policy implications in the final section.

6. Discussion

This research had three objectives. First, to consider the roles of the education and technical departments of PAAs. Secondly, to describe how PAAs' education and technical departments interact with each other and, in turn, with accounting

academia. Third, to identify the dominant institutional logics of technical and education departments of PAAs. The findings are summarised in Table 3.

In this paper the institutional logics theory has been used to analyse the approach by PAAs to academic research; the reasons PAAs fund such research; the type of academic research PAAs choose to fund and how the results of such research are disseminated. There are few studies which apply institutional logics in an accounting context. The two main studies are Thornton, Ocasio, and Lounsbury (2015) and Paisey and Paisey (2017). Thornton et al. (2005) show the rise of commercial logic in professional accounting firms and we find a similar result with the education section of the PAAs.

The research is based on an international data set from senior representatives of PAAs from either end of the world. Between them, the PAAs represented in this study have over half a million members operating across the globe. The research finds that PAAs are very much commercially driven while simultaneously existing to serve the needs of the members. The attempt to serve two masters can create tensions and challenges for PAAs which as CPA Australia discovered can lead to significant member backlash.

Some of the PAAs engaged in this research have also sponsored it and the authors acknowledge their support and their willingness to participate in a completely unencumbered manner. They are all global actors in the accounting arena, often supporting the same international accounting firms. It is somewhat surprising therefore that they have so little motivation to intervene in the protection of the interests of their profession as it is taught and progressed by research in universities while other professional bodies have used their accreditation power to not only protect their discipline but also to ensure it is nurtured and developed. While PAAs do fund academic research there was evidence that the PAAs only valued academic research that was relevant or useful to the accounting practitioner and discounted the academic side of the research. Academic research, especially peer reviewed journal papers which are highly valued in universities, is largely ignored partly because it is regarded as inaccessible and partly because it is not highly prized by a membership that values practical and professional imperatives.

The move to a more commercial logic meant there was the constant need to promote their professional body brand, not just in terms of marketing and public relations exposure for brand recognition and promotion but labelling their brand as a unique international and influential thought leader. This was attributed as one of the main reasons for funding academic research and accrediting their qualifications in higher education institutions.

Academic research is funded by PAAs as it provides them with credibility — a moral legitimisation in the public domain but it does not have any significant influence on the development of the professional curricula. Ironically, pedagogic research is often more highly regarded by the PAAs as it has a direct relevance to improving the teaching and delivery of the professional curricula, while the very same research is less highly regarded by the accounting academic community. However, accounting education research does not feature in the research strategies of the PAAs which serve to reduce the perception of the value and impact of pedagogic research in the academic community. Academic research may also inform policy and provides a branding opportunity for the professional body by suggesting that they are at the cutting edge of thinking in accounting as they compete for student members. Student recruitment is undertaken in a highly competitive market, demonstrating the underlying commercial logic is an important driver for PAAs.

Table 3 Summary of findings.

Objective	Findings
1. Roles of education and research	Education functions charged with growing student and (eventual) member numbers.
functions	Dominance of commercial logics.
	Employers, as consumers of PAAs' talent, determine curriculum, focus on commodity qualifications and 'future proofing' newly-qualified members.
	Research functions often fulfilling a fiduciary role within the PAA, 'advancing the science of accountancy' and
	'public interest concerns' whilst supporting commercial logic of 'increasing brand awareness'
2. Relationship between education and research functions	Generally limited relationship between education and research functions. The division reflects commercial focus of education and fiduciary emphasis of research.
	Education curriculum and student learning not heavily influenced by contemporary accounting knowledge but technical content chosen by employers and traditional learning methods.
3. Institutional logics	Underlying fiduciary logic of the PAA 'in the public interest' contrasted with commercial logic of education to
Mission conflicts	sustain the PAA
Market triggers	i. Benchmarking: ensuring the qualification retains its attractiveness to employers and students.
	ii. Promote the brand, be seen as an influential thought leader.
	iii. React to competition from other providers, maintain and grow market share locally and internationally, attract (scarce) student talent.
	iv. Demonstrate value to membership, ensure 'relevance' of commissioned research.
	v. Meet diverse stakeholder expectations: members, employers, students, and, to a lesser extent, academics
	Dominance of employers in curriculum design.
	vi. Availability of resources, developing the professional curriculum are constrained by limited resources.
Governance forms	General structural separation of education and research reflecting tensions between institutional logics research (fiduciary) and education (commercial)
Changes in dominant logics	Traditional fiduciary logic of PAAs challenged by an emphasis on globalization and commercialism. Dominance of growth expectations and associated targets.

12

The issue of the dissemination of the results of academic research in a manner and format that can be consumed by practitioners continues to be a challenge for the academy. The addition of an impact measure into the research selectivity exercises is being implemented in the UK and under consideration in ANZ. A potential outcome of this would be to provide more recognition and hence incentivise academic researchers to produce more applied research and reports which are of more interest to practitioners. The outcome of such an impact measure will be a closer alignment with the types and forms of research PAAs' support and the PAAs' own commercial logic. The conclusions of this study and some policy implications are discussed in the final section.

7. Conclusions

The paper started with the observation that research in accounting education has long been the poor cousin compared to other areas of accounting research (Marriott et al., 2014; Sangster, 2015). In turn concerns were identified about the adverse impact this will have on the quality of accounting education research in the future (McGuigan, 2015). The aim of this paper was to consider the interaction of the PAAs with the accounting academy in relation to accounting research. Of particular interest was the extent to which the PAAs saw accounting education research being part of such research and the importance and role of such research for the PAAs.

The conclusions to be drawn from this research are many and varied. We identify four. First, PAAs, like the accountancy firms that employ or are owned by so many of their members, are subject to the forces of institutional change driven by tensions between commercial and fiduciary logics (Lander et al., 2013; Suddaby et al., 2015; Thornton et al., 2005). In particular, our work highlights the significance of mission conflicts, market triggers, governance forms and changes in the dominant logics proposed by Thornton et al. (2015).

Second, education and research as the two functions in PAAs that communicate the most with the academic community operate distinct institutional logics. Education emphasises a commercial logic, driven by an organisational desire to grow the membership and maximise employer engagement as a means of attracting new students. Research, adopts a more fiduciary logic where statements about contributing to 'the science of accountancy' and being seen 'to operate in the public interest' exist. However, research itself is sometimes seen as something that promotes the brand suggesting an underlying commercial logic.

Third, the findings point to the continued development of accounting at the creation of hybrid organisations phase. In particular, different PAAs, despite the development of global standards for accounting education, PAAs are each engaged in a quest to differentiate themselves from each other, rather than in as in the past, seeking legitimacy and arguing that some PAAs were more legitimate than others. The growth in the size of PAAs through mergers and organic growth is a result of this competition.

Finally, although links between universities and PAAs are today relatively loose, as Universities aim to produce employable graduates, and accreditation from PAAs convey the attribute of employability. The professional curriculum and research and thought leadership of PAAs is then fundamental to the quality of accounting education and the quality of accounting education research, as accounting education, and allied research, remains fundamentally linked to many matters determined by PAAs, their officers, their members and allied employers.

We acknowledge the research has some limitations. The usual limitations of interview based research apply in terms of the generalisability of the results to PAAs in other jurisdictions. The research considers the views of members of a number of PAAs but excludes those of another important stakeholder group accounting academics. Future research could be conducted with accounting academics to consider their perceptions about the role of PAAs in supporting academic research and the influence of PAAs on the accounting curriculum.

Finally, we identify four policy implications. First, we encourage PAAs to actively promote and engage with accounting education research by including it in their research strategies which would enable the professional curriculum of the PAAs to reflect current academic research in as far as it is relevant to them and the employer community. Second to provide scholarships for early career accounting education researchers would also be a signal as to the value of such research to the academy but also provide much needed support to young academics. Third, they could offer financial support for mentoring programs for such early career researchers as is being piloted in 2018 by the Accounting Education Special Interest Group of the British Accounting and Finance Association and under consideration by the same group in the Accounting and Finance Association of Australia and New Zealand. Finally, we encourage PAAs to consider providing access to data they have on students' performance in the subjects they provide and require students to complete as part of the professional qualification. The data of course would be anonymised and would allow field tests on large data sets to provide valuable information about factors which are important in explaining student success in professional subjects.

Currently, the profession is not benefitting from research-informed teaching or the potential synergies that would exist with an enhanced teaching-research nexus. This would be an opportunity to enact the fiduciary logic of the professional body in a real and engaged approach in curriculum development rather than funding research for branding or commercial gain that is frequently ignored or dismissed by the accounting profession. If there is more support and funding for academics who engage in accounting education research the accounting profession will benefit from a more evidence-based approach to teaching both in the foundation level programs in universities and the professional programmes of the PAAs.

Acknowledgements

The research benefits from financial support from the research committees of the Institute of Chartered Accountants of Scotland and Chartered Accountants Australia and New Zealand.

Appendix. (for review purposes only)

Do you ever ask academics to input into the professional curriculum?

In terms of academic research how broad do you think the profession is to draw the definition of academic research? Is there a link between academic research and the professional curriculum? Does research inform the curriculum?

What would be the motivations for a professional body to include more research into the curriculum or trying to make it more research led or research informed?

Why do professional bodies choose to fund academic research?

How broad is the accounting profession willing to draw the definition of research?

How do you disseminate research findings from the research you have commissioned as a professional body?

References

Abbott, A. (1988). The system of professions: An essay on the division of expert labor. Chicago, IL: University of Chicago Press.

Annisette, M., & Kirkham, L. M. (2007). The advantages of separateness explaining the unusual profession the unusual profession-university link in English Chartered Accountancy. Critical Perspectives on Accounting, 18(1), 1–30.

Armstrong, P. (1985). Changing management control strategies: The role of competition between accountancy and other organisational professions. Accounting, Organizations and Society, 10(2), 129-148.

Basu, S. (2012). How can accounting researchers become more innovative? Accounting Horizons, 26(4), 851-870.

Pathways. (2012). The Pathways commission on accounting higher education: Charting a national strategy for the next generation of accountants Bruce K. B. (C.), William F. E., Leslie A. M.urphy, Judy D. R.ayburn, Melvin T. S.tith, Jerry R. S.trawser. Issues in Accounting Education, 27(3), 595-600.

Carnegie, G. D., Edwards, J. R., & West, B. P. (2003). Understanding the dynamics of the Australian accounting profession: A prosopographical study of the founding members of the incorporated institute of accountants, victoria, 1886 to 1908. Accounting, Auditing & Accountability Journal, 16(5), 790-820. Carr-Saunders, A. M., & Wilson, P. A. (1933). The professions. Oxford: Clarendon Press.

Chartered Accountants Australia New Zealand (CAANZ). (2018). About us accessed 23/4/18 https://www.charteredaccountantsanz.com/about-us/what-wedo/vision-and-strategy.

Clinebell, S. K., & Clinebell, J. L. (2008). The tension in business education between academic rigor and real-world relevance: The role of executive professors. The Academy of Management Learning and Education, 7(1), 99–107.

Creswell, J. W. (1994). Research design: Qualitative & quantitative approaches. Thousand Oaks, CA: Sage Publications, Inc.

Demski, J. S. (2007). Is accounting an academic discipline? Accounting Horizons, 21(2), 159-163.

Dennis, A. (2000). Taking account: Key dates for the profession. Journal of Accountancy, 190(4), 97-103.

Duff, A. (2017). Social mobility and fair access to the accountancy profession in the United Kingdom: Evidence from big four and mid-tier firms, accounting, Auditing Accountability Journal, 30(5), 1082-1110.

Duff, A., & Marriott, N. (2017a). The teaching-research gestalt in accounting: A cluster analytic approach. The British Accounting Review, 49(4), 413-428.

Duff, A., & Marriott, N. (2017b). The teaching-research gestalt: The development of a discipline-based scale. Studies in Higher Education, 42(12), 1–15. Evans, E. (2010). Jurisdictional disputes in accounting: Education or training? In E. Evans, R. Burritt, & J. Guthrie (Eds.), Accounting education at a cross road in

2010. Sydney: ICAA/UniSA.

Financial Reporting Council (FRC). (2017). Key facts and figures. London: FRC. Fletcher, D., Frecknall-Hughes, J., & Williams, S. (2013). Understanding limited liability partnerships in the small and medium-sized business sector. London: Institute of Chartered Accountants in England and Wales.

Friedland, R., & Alford, R. R. (1991). Bringing society back in: Symbols, practices and institutional contradictions. In W. W. Powell, & P. J. DiMaggio (Eds.), The new institutionalism in organizational analysis (2nd. Ed., pp. 232-263). Chicago IL: University of Chicago Press.

Friedson, E. (1994). Professionalism reborn. Chicago IL: University of Chicago Press.

Gammie, E., & Kirkham, L. (2008). Breaking the link with a university education in the creation of a chartered accountant: The ICAS story. The British Accounting Review, 40(4), 356-375.

Goode, W. J. (1961). The librarian: From occupation to profession? The Library Quarterly, 31, 306-320.

Greenwood, E. (1957). Attributes of a profession. Social Work, 2(3), 45-55.

Greenwood, R., Raynard, M., Kodeih, F., Micelotta, E. R., & Lounsbury, M. (2011). Institutional complexity and organizational responses. The Academy of Management Annals, 5(1), 317-371.

Greenwood, R., Suddaby, R., & Hinings, C. R. (2002). Theorizing change: The role of professional associations in the transformation of institutionalized fields. Academy of Management Journal, 45(1), 58-80.

Hancock, P., Marriott, N., & Duff, A. (2017). Research-teaching yin-yang? An empirical study of accounting and finance academics in Australia and New Zealand, Accounting and Finance, 2406–2420.

Hanlon, G. (1994). The commercialization of accountancy: Flexible accumulation and the transformation of the service class. New York: St. Martin's Press. Hanlon, G. (1997). Commercializing the service class and economic restructuring - a response to my critics. Accounting, Organizations and Society, 22(8),

Hopwood, A. G. (2007). Whither accounting research? The Accounting Review, 82(5), 1365-1374.

Hopwood, A. G. (2008). Changing pressures on the research process: On trying to research in an age when curiosity is not enough. European Accounting Review, 17(1), 87-96.

Institute of Public Accountants (IPA). (2018). Annual report 2016-17. Melbourne, Vic: IPA.

Jackall, R. (1988). Moral mazes: The world of corporate managers. International Journal of Politics, Culture, and Society, 1(4), 598-614.

Jones, E. (1995). True and fair: A history of price Waterhouse. London: Hamish Hamilton.

Kaplan, R. S. (2011). Accounting scholarship that advances professional knowledge and practice. The Accounting Review, 86(2), 367–383.

Lampe, J. C., Garcia, A., & Tassin, K. L. (2016). A post-SOX history of U.S. Public accountancy, the history of deprofessionalization in U.S. Public accountancy: Part III. In C. Jeffrey (Ed.), Vol. 20. Research on professional responsibility and ethics in accounting (pp. 3–29).

Lander, M. W., Koene, B. A. S., & Linssen, S. N. (2013). Committed to professionalism: Organizational responses of mid-tier accounting firms to conflicting institutional logics. Accounting, Organizations and Society, 38(2), 130–148.

Larson, M. S. (1977). The rise of professionalism. California: University of California Press.

Lee, T. A. (1995). Shaping the US academic accounting research profession: The American Accounting Association and the social construction of a professional elite. Critical Perspectives on Accounting, 6(2), 241–261.

Macdonald, K. M. (1995). The sociology of the professions. London: Sage.

Marriott, N., Stoner, G., Fogarty, T., & Sangster, A. (2014). Publishing characteristics, geographic dispersion and research traditions of recent international accounting education research. *The British Accounting Review*, 46(3), 264–280.

McGuigan, N. (2015). The impact of journal rankings on Australasian accounting education scholarship – a personal view. Accounting Education: An International Journal, 24(3), 187–207.

Millerson, G. (1964). Qualifying associations. London: Routledge.

Moser, D. V. (2012). Is accounting research stagnate? Accounting Horizons, 26(4), 845–850.

Ocasio, W., & Thornton, P. H. (1999). Institutional logics and the historical contingency of power in organizations: Executive succession in the higher education publishing industry, 1958-1990. *American Journal of Sociology*, 105(3), 801–843.

Oler, D. K., Oler, M. J., & Skousen, C. J. (2010). Characterizing accounting research. Accounting Horizons, 24(4), 635–670.

Paisey, C., & Paisey, N. (2017). The decline of the professionally-qualified accounting academic: Recruitment into the accounting academic community. *Accounting Forum*, 41, 57–76.

Previts, G. J. (1985). The accounting profession, years of trial: 1969-1980. The Accounting Review, 60(2), 335-336.

Ramirez, C. (2009). Constructing the governable small practitioner: The changing nature of professional bodies and the management of professional accountants' identities in the UK, Accounting, Organizations and Society, 34(3–4), 381–408.

Ratzinger-Sakel, N. V. S., & Gray, G. L. (2015). Moving toward a learned profession and purposeful integration: Quantifying the gap between the academic and practice communities in auditing and identifying new research opportunities. *Journal of Accounting Literature*, 35(1), 77–103.

Rebele, J. E., St, K., & Pierre, E. (2015). Stagnation in accounting education research. Journal of Accounting Education, 33(2), 128-137.

Report, C.harles (1970). Tomorrow's CA in the making. Edinburgh: Institute of Chartered Accountants of Scotland (ICAS).

Richardson, A. (1988). Accounting knowledge and professional privilege. Accounting, Organizations and Society, 13(4), 381-396.

Richardson, A. J. (1997). Social closure in dynamic markets: The incomplete professional project in accountancy. *Critical Perspectives on Accounting*, 8(6), 635–653.

Richardson, A. J. (2017). Merging the profession: A social network analysis of the consolidation of the accounting profession in Canada. Accounting Perspectives, 16(2), 83–104.

Richardson, A. (2018). In L. Goretzki, & E. Strauss (Eds.), The relationship between management accounting and financial accounting as professions and technologies of practice, the Role of the Management Accountant: Local Variations and Global Influences. London: Routledge.

Saldaňa, J. (2009). The coding manual for qualitative researchers. Thousand Oaks, CA: Sage.

Sangster, A. (2015). You cannot judge a book by its cover: The problems with journal rankings. Accounting Education: An International Journal, 24(3), 175–186.

Silverman, D. (2014). *Interpreting qualitative data* (5th ed.). London: Sage.

Suddaby, R., Cooper, D. L., & Greenwood, R. (2007). Transnational regulation of professional services: Governance dynamics of field level organizational change. *Accounting, Organizations and Society, 32*(4–5), 333–362.

Suddaby, R., Gendron, Y., & Lam, H. (2009). The organizational context of professionalism in accounting. *Accounting. Organizations and Society*, 34(3–4), 409–427.

Suddaby, R., & Greenwood, R. (2005). Rhetorical strategies of legitimacy. Administrative Science Quarterly, 50(1), 5-67.

Suddaby, R., Saxton, G. D., & Gunz, S. (2015). Twittering change: The institutional work of domain change in accounting expertise, *Accounting. Organizations and Society, 45*(1), 52–68.

Thornton, P. H., Jones, C., & Kury, K. (2005). Institutional logics and institutional change in organizations: Transformation in accounting, architecture and publishing. *Research in the Sociology of Organizations*, 23, 125–170.

Thornton, P. H., & Ocasio, W. (1999). Institutional logics and the historical contingency of power in organizations: Executive succession in the higher education publishing industry, 1958–1990. *American Journal of Sociology*, 105(3), 801–843.

Thornton, P. H., Ocasio, W., & Lounsbury, M. (2012). The institutional logics perspective: A new approach to culture, structure and process (Oxford Scholarship Online).

Thornton, P. H., Ocasio, W., & Lounsbury, M. (2015). The institutional logics perspective. In R. A. Scott, & M. C. Buchmann (Eds.), *Emerging trends in the social and behavioral sciences: An interdisciplinary, searchable, and linkable resource* (pp. 1–22). Wiley Online Library.

Van Wyhe, G. (2007). A history of U.S. Higher education in accounting, Part I: Situating accounting within the academy. *Issues in Accounting Education*, 22(2), 163–181.

Walker, S. P. (1995). The genesis of professional organization in Scotland: A contextual analysis. Accounting, Organizations and Society, 20(4), 285-310.

Walker, S. P. (1999). Introduction to histories of accounting professionalism. Accounting Business and Financial History, 9(1), 1-6.

Walker, S. P. (2004). The genesis of professional organization in English Accountancy. Accounting, Organizations and Society, 29(2), 127-156.

Wilkerson, J. E., Jr. (2010). Accounting educators as the accounting profession's trustees: Lessons from a study of peer professions. *Issues in Accounting Education*, 25(1), 1–13.

Zeff, S. A. (2003a). How the US accounting profession got where it is today: Part I. Accounting Horizons, 17(3), 189–205.

Zeff, S. A. (2003b). How the US accounting profession got where it is today: Part II. Accounting Horizons, 17(4), 267-286.